

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Duodu, Emmanuel; Baidoo, Samuel Tawiah; Yusif, Hadrat; Frimpong, Prince Boakye

### Article

# Money supply, budget deficit and inflation dynamics in Ghana: An empirical investigation

**Cogent Business & Management** 

### Provided in Cooperation with:

Taylor & Francis Group

*Suggested Citation:* Duodu, Emmanuel; Baidoo, Samuel Tawiah; Yusif, Hadrat; Frimpong, Prince Boakye (2022) : Money supply, budget deficit and inflation dynamics in Ghana: An empirical investigation, Cogent Business & Management, ISSN 2331-1975, Taylor & Francis, Abingdon, Vol. 9, Iss. 1, pp. 1-23, https://doi.org/10.1080/22211075.2022.2042810

https://doi.org/10.1080/23311975.2022.2043810

This Version is available at: https://hdl.handle.net/10419/288559

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.









**Cogent Business & Management** 

ISSN: (Print) (Online) Journal homepage: www.tandfonline.com/journals/oabm20

# Money supply, budget deficit and inflation dynamics in Ghana: An empirical investigation

Emmanuel Duodu, Samuel Tawiah Baidoo, Hadrat Yusif & Prince Boakye Frimpong

**To cite this article:** Emmanuel Duodu, Samuel Tawiah Baidoo, Hadrat Yusif & Prince Boakye Frimpong (2022) Money supply, budget deficit and inflation dynamics in Ghana: An empirical investigation, Cogent Business & Management, 9:1, 2043810, DOI: 10.1080/23311975.2022.2043810

To link to this article: https://doi.org/10.1080/23311975.2022.2043810

© 2022 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.



6

Published online: 07 Mar 2022.

Submit your article to this journal 🕑





View related articles 🗹



View Crossmark data 🗹



Citing articles: 6 View citing articles





Received: 05 April 2021 Accepted: 12 February 2022

\*Corresponding author: Samuel Tawiah Baidoo, Department of Economics, Kwame Nkrumah University of Science and Technology, Private Mail Bag, Kumasi Ghana E-mail: samueltawiahbaidoo@yahoo. com

Reviewing editor: David McMillan, University of Stirling, UNITED KINGDOM

Additional information is available at the end of the article

### **BANKING & FINANCE | RESEARCH ARTICLE**

# Money supply, budget deficit and inflation dynamics in Ghana: An empirical investigation

Emmanuel Duodu<sup>1,2</sup>, Samuel Tawiah Baidoo<sup>1,3</sup>\*, Hadrat Yusif<sup>1</sup> and Prince Boakye Frimpong<sup>1</sup>

**Abstract:** The paper investigates the long run dynamics of money supply, budget deficit and inflation in Ghana. It also tests the validity of the classical, monetary and fiscal theories of price level within the vector error correction framework. Using quarterly data from 1999Q1 to 2019Q4, the paper employs Granger causality test and the vector error correction model (VECM) for the analysis. The results from the VECM show that budget deficit has a significant positive effect on inflation while money supply negatively affect it. By contrast, inflation exerts a positive and negative effect on budget deficit and money supply, respectively. The results from the impulse response function also indicate that inflation responds more positively to budget deficit shocks. However, it tends to respond negatively to money supply (M2) shocks. Also, budget deficit responds positively (negatively) to inflation (money supply [M2]) shocks. Furthermore, money supply responds positively (negatively) to

#### ABOUT THE AUTHOR

Emmanuel Duodu is currently a Master of Science student at Department of Fundamentals of Economic Analysis, University of Alicante, Alicante-Spain. His research interests include macroeconomics, international economics, monetary economics and public economics.

Samuel Tawiah Baidoo is a PhD candidate at the Department of Economics, Kwame Nkrumah University of Science and Technology (KNUST), Kumasi-Ghana, and a lecturer at the Department of Accounting and Finance, School of Business, Christian Service University College, Kumasi-Ghana. His research interests include microeconomics, monetary economics, international economics and economic policy analysis.

Hadrat Yusif is a Professor at the Department of Economics, Kwame Nkrumah University of Science and Technology (KNUST), Kumasi-Ghana. He holds PhD in Economics from the National University of Malaysia. He is specialised in economics of education and monetary economics.

Prince Boakye Frimpong holds a PhD in Economics from the University of Milan, Italy. He is currently a senior lecturer at the Department of Economics, KNUST. His areas of specialisation are Applied Econometrics and general Applied Microeconomics.

#### PUBLIC INTEREST STATEMENT

The dynamics of money supply, budget deficit and inflation continue to receive attention from both theoretical and empirical perspectives considering the negative repercussions inflation is likely to have on economies as well as the livelihood and welfare of the citizenry. Several studies have investigated the phenomenon in the Ghanaian context but fail to incorporate the key assumption regarding the exogenous property of money supply and budget deficit in explaining inflation. Therefore, the present study contributes to knowledge by incorporating the exogeneity test into the analysis to establish the exogenous property of money supply and budget deficit. This helps to completely ascertain the dynamics of money supply, budget deficit and inflation. The results show that budget deficit has a significant positive effect on inflation while money supply negatively affect it. Based on the weak exogeneity test, the result favours the fiscal theory of the price level in explaining the nexus between money supply, budget deficit and inflation in Ghana.





 ${\small ©}$  2022 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

budget deficit (inflation) shocks. Based on the weak exogeneity test, the result favours the fiscal theory of the price level in explaining the nexus between money supply, budget deficit and inflation in Ghana. A corollary of our results is that a reduction in government expenditure coupled with restrictive bureaucratic nature of government officials have the tendency of ensuring favourable and stable inflation in Ghana.

## Subjects: Economics and Development; Economics; Finance; Business; Management and Accounting

#### Keywords: Budget deficit; Ghana; inflation; money supply; weak exogeneity test

#### 1. Introduction

The dynamics of money supply, budget (fiscal) deficit and inflation continue to receive attention from both theoretical and empirical perspectives considering the negative repercussions inflation is likely to have on economies (Adom et al., 2018), as well as the livelihood and welfare of citizens. Inflation is perceived to be a key macroeconomic indicator as it influences major decisions such as investment, consumption and savings among others. Inflation also renders the execution of some major policies or projects unsuccessful as it affects the budget allocation, and this ultimately retards the progress of an economy. In the Ghanaian context, successive governments have attempted to maintain a single digit as well as stable inflation rate with the aim of improving the wellbeing of citizenry and enhancing savings and investment decisions. However, these efforts have proven futile as inflation rate in the economy continues to fluctuate and remains double digit. For instance, inflation rate in Ghana averaged 22.40, 13.18, 11.50 and 12.91 percent for the periods 1999-2003, 2004-2008, 2009-2013 and 2014-2019, respectively (World Bank, 2019). Furthermore, the average inflation rate from 1999 to 2019 was 15.02%. This clearly indicates that inflation has not been stable and therefore could adversely affect major economic decisions. Given this, it is therefore prudent to ascertain the determinants of inflation in the Ghanaian economy for effective policy purposes; hence, the need for the present study.

From theoretical standpoint, the Classicalists and Monetarists have based their arguments regarding the nexus between money supply, budget deficit and inflation on what is known as the quantity theory of money, whereas that of the fiscal theory of the price level (FTPL) has been based on the quantity theory of government debt. These notwithstanding, the Keynesians and the Structuralists have argued differently about the nexus and dynamics among these macroeconomic variables. The Keynesians believe that inflation is caused by excessive demand oppressions in an economy, whereas the Structuralists argue that inflation in underdeveloped and developing countries is due to structural (economic and social) characteristics of an economy.

The diverse views among these school of thoughts have drawn the attention of researchers and policymakers in various economies to empirically test the dynamics proposed by these theories to ensure that appropriate measures or policies are put in place to curb inflation. In view of that, many scholars have extensively explored the relationship between money supply, budget deficit and inflation in both developing and developed countries with mixed results (Adom et al., 2015; Adu & Marbuah, 2011; Ahiabor, 2013; Anantha Ramu, 2014; Boamah, 2019; Chiaraah & Nkegbe, 2014; Gyebi & Boafo, 2013; Kovanen, 2011; Lin & Chu, 2013; Nasir et al., 2020a, 2020b; Nguyen, 2015).

This paper contributes to knowledge and extant literature by way of validating existing studies on the nexus between money supply, budget deficit and inflation in the Ghanaian context. This study also fills the lacuna that has been created regarding the links between these variables by considering the long run relationship between money supply, budget deficit and inflation. The reason for focusing primarily on the long run relationship is because the short run relationship may not be sufficient (though necessary) for effective policy discourse and therefore could render policies unsuccessful in an economy. Furthermore, the present study tests the validity of classical theory, monetarist theory and the fiscal theory of the price level within the context of Ghana.

Specifically on Ghana, studies (see: Sowa, 1994; Ghartey, 2001; Bawumia and Abradu-Otoo, 2003; Adu & Marbuah, 2011; Kovanen, 2011; Ahiabor, 2013; Gyebi & Boafo, 2013; Chiaraah & Nkeabe, 2014; Adom et al., 2015; Ibn Boamah, 2019) have examined the causal links between money supply, budget deficit and inflation. Results from these studies are mixed and have mostly considered the nexus using annual data, thereby ignoring how budget deficit and money supply influence inflation, and vice versa using quarterly data. Considering high-frequency data (quarterly data) is very important because it is likely to change the dynamics that exist among budget deficit, money supply and inflation and therefore could have a significant impact on policy implementation and hence, the present study becomes very essential. Another key contribution of the present study is the exogeneity test it incorporates in the analysis, which has been ignored in all previous studies (Sowa, 1994; Ghartey, 2001; Bawumia and Abradu-Otoo, 2003; Adu & Marbuah, 2011; Kovanen, 2011; Ahiabor, 2013; Gyebi & Boafo, 2013; Chiaraah & Nkeabe, 2014; Adom et al., 2015; Ibn Boamah, 2019). In understanding the relationship between money supply, budget deficit and inflation, an exogeneity analysis is very key in explaining the theoretical links. The neglect of the exogeneity property of money supply and budget (fiscal) deficit contradicts theoretical views on money supply and budget (fiscal) deficit. For instance, according to the classical and monetarist theories, money supply is exogenously determined by the monetary authorities, and hence inflation is everywhere a monetary phenomenon. By contrast, fiscal deficit, which is perceived to cause inflation in the long run, is assumed to be exogenous by the fiscal theory of the price level and the monetarist hypothesis. Thus, a test of exogeneity of these two variables in explaining inflation dynamics is crucial to understanding precisely the theoretical arguments espoused in the literature. More tersely, the paper seeks to test (using the weak exogeneity test) which of the theories is more relevant in explaining inflation dynamics in Ghana.

We find that budget deficit has a significant positive effect on inflation while money supply negatively affects it. By contrast, inflation exerts a positive and negative effect on budget deficit and money supply, respectively. Based on the weak exogeneity test, the result favours the fiscal theory of the price level in explaining the nexus between money supply, budget deficit and inflation in Ghana.

The rest of the paper is structured as follows: Section 2 reviews the theoretical and empirical literature, whiles Section 3 presents the data and empirical strategy used in the study. Section 4 presents the empirical results and the discussion thereof. Section 5 concludes the paper with some policy implications.

#### 2. Theoretical and empirical literature review

Several theories related to the dynamics of money supply, budget deficit and inflation have emerged in monetary economics. This section provides a brief review of theories as well as earlier studies on money supply, budget deficit and inflation nexus. With respect to the theoretical underpinning, the classical theory, the monetarist hypothesis (MH) and the fiscal theory of the price level (FTPL), the Keynesian approach and the Structuralist theory of inflation are reviewed.

The classical theory, based on the quantity theory of money, argues that the quantity of nominal money supply, which is exogenously determined is sorely responsible for changes in the general price level (inflation) in an economy. Thus, changes (increase and decrease) in the nominal money supply directly leads to a change in the general price level at an equal proportion. The theory further argues that the real sector operates independently from the monetary sector, and for that matter, real wages are determined in the real sector whiles changes in nominal money supply

determine the nominal wages. Therefore, real output in an economy is not influenced by an increase in nominal money supply but general price level (inflation) rises when nominal money supply increases. The classical theory therefore views inflation as a monetary phenomenon.

The monetarist hypothesis (theory), just like the classical theory, is based on the quantity theory of money, which requires that the price level is determined (controlled) by the nominal money supply. The proponents argue that the general price level is exclusively determined as the price that ensures that the desired level of real balances equals the purchasing power of the nominal money supply at any given level of nominal money supply, which is determined (exogenously) by the monetary authority. The implication is that deviation of nominal money supply from the desired real balances (given any price level) will channel into changes in the price level. Notwithstanding, the monetarists further posit that budget deficit is the cause of inflation through money supply. The reason is that a budget deficit financed by monetary authority (Central Bank) through seigniorage (printing of money by the central bank) or financing government expenditure through the open market operation (purchase of government interest-bearing securities) changes the nominal money supply in an economy and hence changes in the general price level. Given this, Friedman (1963) argues that inflation is always and everywhere a monetary phenomenon.

The fiscal theory of the price level (FTPL) also known as the quantity theory of government debt analyzes how fiscal (budget) deficit feeds through general price level in different mechanism other than that of the monetarist approach. The FTPL considers the government's intertemporal budget constraint (GBC) as an instrument that links both fiscal and monetary policies. According to the FTPL, the GBC is at equilibrium when the discounted value of the government's future primary surplus (which includes seigniorage as a revenue source) is greater than (equal to) the current nominal value of the government (public) debt, which considers the monetary base. The proponents opine that the discount rate is measured by the ratio of real interest rate to the growth rate of the economy. The FTPL assumes that the future path of revenues and primary expenditures is determined exogenously by fiscal authority. The theory further argues that, at a given discount rate, the price level will rise to equilibrate the GBC condition anytime the discounted value of primary surplus is lower than the value of nominal public debt. Therefore, price is the only adjustment variable to maintain equilibrium condition in the GBC.

To understand how the price level is affected by fiscal policy, Woodford (1995) suggests that a positive and exogenous price shock reduces the value of government debt (liabilities) owed to private individuals who have purchased or invested in government securities which in turn lowers their wealth as well as demand for goods. The FTPL theory postulates that, when this happens, the individual's expectations with respect to the sustainability of fiscal policy will generate similar wealth-effect. If the market recognizes a negative perception about the sustainability of public finances (when discounted value of government primary surplus deviate from the nominal value of government liabilities), such negative perception will trigger an increase in the level of price to a higher level required to equalize the GBC. This higher price lowers the value of private assets, which generates the abovementioned wealth-effect. Therefore, higher government debt (liabilities) generates higher distortion, and hence, higher prices are required to restore the GBC. The implication is that budget deficit causing long run inflation with money supply playing no role may establish a strong backing for the FTPL as indicated by Lozano-Espitia and Lozano-Espitia (2008).

On the contrary, the Keynesian approach to determination of inflation is subjected to excessive demand dominations, which assumes that the economy is at full employment. Keynes argues that firms generate more profit at a fixed nominal wages when there exists excess demand at full employment level. As result, firms demand for labour increases with the aim of meeting the growing demand in the economy, which in turn leads to higher wages paid by firms. As documented by Kotwal (1987) and Frisch (1989)), the higher wages increase the general price level as

cost of production increases and hence inflation arise. The structuralist theory of inflation on the other hand attributes inflation in underdeveloped and developing countries to the structural characteristics (economic and social structure) in the process of economic growth but not mone-tary factors as argued by the classical and the monetarist economists. Given the idea of structuralism, early contributors like Noyola (1956), Sunkel (1958), Olivera (1964) and Chenery (1975) argued that aggregate demand in an economy increases in periods of rapid economic growth and development but the increase in the aggregate supply does not match the increase in aggregate demand. These contributors therefore posit that the gap is because of structural bottlenecks. Arndt (1985) indicates that the gap between aggregate demand and supply caused by the structural bottlenecks results in inflationary pressure in the economy. Thus, the overall price level in the economy increases due to shortage of goods as the economy grows and develops.

The conflicting propositions by the aforementioned theories have therefore inspired many researchers to empirically analyze the causal effect of money supply, budget deficit and inflation in both developing and developed world.

Specifically, on Ghana, studies have been conducted to analyze the dynamics among money supply, budget deficit and inflation. For instance, Sowa (1994) examined the relationship between fiscal deficits, output growth and inflation targets for the period 1965 to 1991 using the error correction model (ECM) as estimation technique. The results revealed that nominal money (M0) and real income have a significant positive impact on inflation, whereas exchange rate tend to have a positive significant impact on inflation. The study further indicated that, for periods with consistent fiscal deficit or policy (inconsistent fiscal deficit), inflation tends to be within target (above target).

Also, Ghartey (2001) investigated macroeconomic instability and inflationary financing nexus using quarterly time series data covering the period 1970 to 1992. Employing the pair-wise Granger causality test and vector error correction model (VECM) for the analysis, the study showed that monetary base and currency ratio cause inflation and real output growth and inflation also cause exchange rate growth. Real output growth is revealed to have a bidirectional causal relationship with money supply growth, monetary base, currency ratio, budget deficit as a percentage of GDP, and inflation. Similarly, using annual time series data spanning 1983 to 1999, Bawumia and Abradu-Otoo (2003) explored the relationship between monetary growth, exchange rates and inflation. The results from the error correction model showed that money supply (M2+) and exchange rate have a significant positive relationship with inflation, whereas the effect of real income on inflation is revealed to be negative and significant.

Using the ordinary least squares and generalized method of moments estimation techniques, Kovanen (2011) investigated whether money matter for inflation using quarterly time series data spanning 1990 to 2009. The study revealed that inflation gap and real output gap have a positive and significant effect on inflation. Real money gap and nominal money gap are also found to have insignificant negative effect on inflation in both estimation techniques. Currency depreciation is also found to have significant negative (significant positive) effect on inflation in four quarters (eight quarters) in both the OLS and GMM estimators. Also, Adu and Marbuah (2011) employed the autoregressive distributed lag (ARDL) model to examine the determinants of inflation using annual time series data over the period 1960 to 2009. The study showed that money supply (M1, M2 and M3) has a significant positive influence on inflation in both the long and short run. The relationship between fiscal deficit and inflation is revealed to be insignificant in the long run but positive and significant in the short run. Exchange rate is found to exert significant negative (insignificant positive) effect in the long run (short run). The study further showed that, while interest rate impacted positively on inflation, real output is found to have a significant negative effect on inflation in both the long- and short-run.

In a related study, Adom et al. (2015) analyzed inflation dynamics using annual time series data covering the period 1960 to 2012. The study employed the fully modified ordinary least squares (FMOLS) estimation technique for the analysis. The study concluded that money supply exerts a positive significant impact on inflation, whereas fiscal deficit has insignificant effect on inflation. Crude oil price and interest rate are found to have positive and significant effect on inflation. Food production index is revealed to impact negatively on inflation. The results further indicated that output growth and exchange rate have insignificant effect on inflation. Similarly, Adjei (2018) explored the monetarists' theory on inflation determinants using annual time series data spanning 1965 to 2012. Applying the autoregressive distributed lag (ARDL) model as the estimation techniaue, the study found that broad money (M2) growth and broad money as a percentage of GDP have a significant positive relationship with inflation in the long run. In the short-run, only broad money growth is revealed to have significant positive effect on inflation and broad money as a percentage of GDP impacts negatively on inflation. The results further showed that import of goods and services have insignificant negative (significant positive) effect on inflation in the long run (short run), whereas GDP per capital growth exert a negative and significant (positive and significant) impact on inflation in the long run (short run). Domestic credit to the private sector is found to have insignificant effect on inflation in both the long and short run.

Furthermore, Boamah (2019) investigated inflation dynamics using annual time series data from 1972 to 2016, and the ordinary least squares (OLS) and error correction model (ECM) are applied to the dataset. The results from the OLS revealed that money supply (M2), real effective exchange rate, foreign price and nominal interest rate exert a significant positive impact on inflation. The results from the ECM also showed that real effective exchange rate, foreign price and nominal interest rate a significant, whereas money supply (real income per capita) is found to exert insignificant positive (significant negative) effect on inflation.

In Colombia, Lozano-Espitia and Lozano-Espitia (2008) examined the relationship between budget deficit, money growth and inflation in Colombia. The study employs quarterly (annual) time series data from the period 1982Q1 to 2007Q4 (1955 to 2007) and applies the vector error correction model to the dataset. The results from the quarterly data showed that money growth (M0, M1 and M3) has a positive and significant impact on inflation, and inflation also affects money growth positively. The results further indicate that budget deficit has a significant positive effect on money growth (M1). The result from the annual data is, however, not different from the quarterly data except for budget deficit having insignificant effect on both inflation and money growth (M0, M1 and M3).

Employing time series data from the period 1967 to 2010, Ndanshau (2012) investigated the nexus between budget deficits, money supply and inflation in Tanzania. The pair-wise Granger causality test and vector error correction model (VECM) are used for the analysis. The study showed that inflation Granger causes budget deficit, budget deficit Granger causes money supply (M0), money supply (M1) Granger causes inflation, inflation Granger causes money supply (M0) and inflation Granger causes budget deficit when deflated with monetary base (M0). The VECM results revealed that budget deficit has insignificant positive effect on inflation, but inflation exerts significant negative effect on budget deficit. Policy regime is also revealed to exert a positive and significant effect on both inflation and budget deficit.

Similarly, Lin and Chu (2013) employed the dynamic panel quantile regression (DPQR) model under the autoregressive distributional lag (ARDL) specification to examine deficit-inflation relationship in 91 countries (OCED and non OCED) for the period 1960 to 2006. The results showed that current and lagged deficit have positive and significant impact on inflation. DPQR results with additional explanatory variables revealed that deficit and oil price inflation have a positive and significant effect on inflation, whereas trade openness has significant negative relationship with inflation. The effect of real GDP growth is insignificant. With regard to separate OCED and non-OCED country analysis, the results showed that deficit, real GDP growth and oil price exert a significant positive impact on inflation, whereas the effect of trade openness is negative in OCED countries. For the non-OCED countries, the study shows that deficit and oil price have significant positive effect on inflation, but the effect of real GDP growth and trade openness is negative and significant. In a related study, Anantha Ramu (2014) investigated the relationship between fiscal deficit and inflation in India using time series data from the period 1980–81 to 2011–12. The results from the autoregressive distributed lag (ARDL) model indicated that gross fiscal deficit, interest rate and oil import price have a significant positive effect on inflation, whereas the effect of exchange rate and GDP growth rate are negative and significant. Money supply growth rate is found to have insignificant positive impact on inflation.

Also, Nguyen (2015) examined the effect of fiscal deficit and money supply (M2) on inflation in selected economies (Bangladesh, Cambodia, Indonesia, Malaysia, Pakistan, Philippines, Sri Lanka, Thailand and Vietnam) of Asia. Using time series data from the period 1985 to 2012, the study applies the pooled mean group (PMG) estimation-based error correction model and the panel differenced GMM estimation techniques. The results from the PMG indicated that fiscal deficit and money supply (M2) have a significant positive effect on inflation in long run, but in the short run, only money supply has significant negative effect on inflation, and the effect of fiscal deficit is insignificant. The outcome from the GMM also showed that fiscal deficit has significant positive effect on inflation, whereas the effect of money supply is revealed to be insignificant. Also, whereas government expenditure and interest rate exert significant positive effect on inflation, the effect of real GDP per capita, exchange rate and trade openness are insignificant in both PMG and GMM estimators.

More recently, Nasir et al. (2020a) examined inflation expectations in the face of oil shocks for New Zealand and United Kingdom from the period January 1984 to June 2018. The results from the non-linear autoregressive distributed lag (NARDL) model indicated that real effective exchange rate has a significant negative relationship with inflation expectations in both the short- and long-run for both countries. The results further revealed that inflation, real effective exchange rate, money supply, output growth, unemployment and fiscal deficit/ surplus have significant implications for inflation expectations for the two countries. Further, Nasir et al. (2020b) investigated the exchange rate pass-through and management of inflation expectations for Czech Republic using the NARDL. The outcome of the study showed that real effective exchange rate has a significant negative relationship with inflation expectations in both the short- and long-run. However, the relationship between inflation expectations and oil price shocks is positive but insignificant in both periods. Fiscal stance and money supply are also revealed to have insignificant negative relationship with inflation expectations in both periods.

Indisputably, evidence from past studies on Ghana and other parts of the world have revealed mixed (inconclusive) results regarding budget deficit, money supply and inflation dynamics; hence, further research on the subject matter becomes imperative as far as validity of previous outcomes is concerned. Again, it is evident that studies on Ghana that use quarterly data for the analysis rarely exist. Mention can only be made to studies by Ghartey (2001) and Kovanen (2011). Given the fact that high frequency data (quarterly) has the tendency of changing the dynamics that exist among budget deficit, money supply and inflation, as well as having significant impact on policy implementation, it is therefore worthwhile for studies to focus on high frequency data in such analysis, and the present study seeks to contribute to literature in that direction. It is also revealed from the empirical review that none of the studies, especially those on Ghana, incorporates the exogeneity test in the analysis, and this renders these studies incomplete to a greater extent. This is so because, the major issue about the exogenous nature of the determinants of inflation such as money supply and budget deficit, which theories postulate, is ignored. The present study therefore incorporates this major test in the analysis to fully ascertain the validity of the theories that seek to explain the nexus between budget deficit, money supply and inflation, and this is a key contribution of this study to existing literature.

#### 3. Methodology and data

This section presents the theoretical and methodological framework, as well as data and estimation technique the study employs. The section is divided into three parts. The first part presents the theoretical framework and model specification. The second part focuses on the estimation strategy, whereas the third part presents the data and variable description.

#### 3.1. Theoretical framework and model specification

The study adopts a theoretical model of inflation with the aim of identifying the important factors that influence inflation in a small open economy like Ghana. Inflation, measured as the percentage change in the overall (general) price level of goods and services, is defined as the persistent (sustained) increase in the overall price level of goods and services in an economy. The model assumes a small open economy, where the general price level is given as the weighted average of the price of tradable goods ( $P^{T}$ ) and the price of non-tradable goods ( $P^{NT}$ ), which is expressed in equation (1).

$$log1muP_{t} = \gamma logP^{T} + (1 - \gamma) logP^{NT}$$
<sup>(1)</sup>

where  $\gamma$  is the weight of tradable goods and lies within zero and one (0 <  $\gamma$  < 1).

In the model of small open economy, the world market determines the price of tradable goods  $(P^{T})$ , which depends on foreign price  $(P^{f})$  assumed as given and exchange rate (ER). By incorporating the purchasing power parity, price of tradable good can then be express as follows:

$$P^{\mathsf{T}} = P^{\mathsf{f}} E R \tag{2}$$

Taking logarithm of equation (2) then gives equation (3).

$$\log P^{T} = \log P^{f} + \log ER \tag{3}$$

Since a small open economy's influence in the world market is completely insignificant, foreign price is treated as given, which without loss of generality is normalized to one, and hence exchange rate becomes the only determinant of the price of tradable goods, which is specified in equation (4).

$$log P^{T} = log ER \tag{4}$$

The implication of equation (4) is that changes in exchange rate may cause the price of tradable goods to change by the same proportion, especially if the economy is import dependent. Equation (4) also indicates that exchange rate plays very crucial role in determining domestic inflation (as depreciation leads to domestic inflation), hence a partial exchange rate pass-through effect on domestic prices (see: Amoah & Aziakpono, 2018).

The model further assumes that the domestic money market on the other hand determines the price of non-tradable goods, where the demand for non-tradable goods is assumed, for simplicity, to move together with the overall demand of the economy. The implication is that the money market equilibrium (which requires equalization between real money supply and real money balances) determines the price of non-tradable goods. This is expressed in equation (5).

$$P^{\rm NT} = \delta(\log M^{\rm s} - \log M^{\rm d}) \tag{5}$$

where  $\delta$  is a scale factor denoting the relationship between demand for non-tradable goods and aggregate demand of the economy. The demand for real money balances is assumed to depend on real income and inflationary expectations. Interest rate, also being the opportunity cost of money, determines the demand for money. Therefore, the demand function for real money balances can be represented as follows.

$$M^d = f(Y, \pi^E, I) \tag{6}$$

where Y,  $\pi^E$  and I denote real income, expected inflation rate and nominal interest rate, respectively. Demand for money theory assumes that there exists a positive relationship between real income, expected inflation and demand for real money, whereas on the other hand, the relationship between real money demand and interest rate is assumed to be negative. Based on Ubide (1997), Laryea and Sumaila (2001), Adu and Marbuah (2011), and Adom et al. (2015), the study specifies the following inflationary expectation equation.

$$\pi^{E} = \tau[L(\pi_{t})] + (1 - \tau)\Delta logP_{t-1}$$
<sup>(7)</sup>

where  $L(\pi_t)$  denotes a distributed lag learning process for economic agents (consumer and producer). Without loss of generality, the expectations of these agents are assumed to be fully backward looking, which means that the coefficient ( $\tau$ ) of the distributed lag learning process equals zero. Following this, equation (7) can be reduced to equation (8).

$$\pi^{E} = \Delta \log P_{t-1} \tag{8}$$

Substituting equation (8) into equation (6) generates equation (9).

$$M^d = f(Y, P_{t-1}, I)$$
 (9)

By further substituting equation (9) into equation (5) also gives equation (10).

$$P^{NT} = \delta[\log M^{s} - \log f(Y, P_{t-1}, I)]$$
<sup>(10)</sup>

Substituting both equations (10) and (4) into equation (1) and rearranging then gives the final inflation equation as:

$$\log P_{t} = \gamma \log ER + (1 - \gamma) \log [M^{s} - f(Y, P_{t-1}, I)]$$
(11)

From equation (11), the general price level of goods and services in a small open economy can, therefore, be generally expressed in equation (12).

$$P_{t} = f(Y_{t}, ER_{t}, M_{s}^{s}, I_{t}, P_{t-1})$$
(12)

It is observed from equation (12) that, inflation in a small open economy is influence by real income, exchange rate, money supply, nominal interest rate and previous year's price level (inflation).

The general specification of the overall price level of goods and services [equation (12)] is then modified to capture other variables such as budget deficit, which has the potential of influencing inflation. For instance, financing budget deficit through seigniorage (printing money) or purchases of government securities leads to temporal disequilibrium in the money market and could therefore leads to inflation.

The modified equation (12) for estimation is therefore specified in equation (13).

$$P_t = f(Y_t, ER_t, M_t^s, I_t, BD_t)$$
(13)

where  $P_t$ ,  $Y_t$ ,  $ER_t$ ,  $M_t^s$ ,  $I_t$  and  $BD_t$  represent the overall price level (Inflation), real income, money supply [measured by narrow (M1) and broad (M2) money], nominal interest rate and budget (fiscal) deficit, respectively, and t denotes the time trend.

The estimable form of equation (13) is given by equation (14).

$$\ln P_t = \alpha_0 + \beta_1 \ln M_t^s + \beta_2 \ln BD_t + \beta_3 \ln ER_t + \beta_4 \ln Y_t + \beta_5 \ln I_t + \varepsilon_t$$
(14)

where the variables in equation (14) are as explained earlier.  $M_t^s$  is measured by M1 and M2, which represent narrow and broad money supply, respectively,  $\alpha_0$  and  $\varepsilon_t$  are the constant and the

stochastic error terms, respectively, such that the error term is normally distributed with a mean of zero and a constant variance [ $\varepsilon_t \sim N(0, \sigma^2)$ ]. Again, the  $\beta$ 's (1, 2, 3, ..., 5) are the respective coefficients of the variables to be estimated, and ln denotes the natural logarithm. It must be emphasized that equation (14) is estimated twice. In the first and second cases, the equation is estimated using narrow money supply (M1) and broad money supply (M2), respectively. In each estimation, inflation is normalized first, and this is followed by money supply normalization and finally budget deficit normalization. The motivation for this normalization is to see how each variable (budget deficit, money supply and inflation) influences the other. The choice of the variables for the present study is influenced by existing works (Ghartey, 2001; Bawumia and Abradu-Otoo, 2003; Lozano-Espitia & Lozano-Espitia, 2008; Adu & Marbuah, 2011; Kovanen, 2011; Ahiabor, 2013; Gyebi & Boafo, 2013; Lin & Chu, 2013; Anantha Ramu, 2014; Chiaraah & Nkegbe, 2014; Adom et al., 2015; Nguyen, 2015; Ibn Boamah, 2019; Nasir et al., 2020a, 2020b).

#### 3.2. Estimation strategy

The study adopts the vector error correction model (VECM) propounded by Johansen (1988) and Granger causality test (Granger, 1969) to test the hypothesis regarding the relationship between money supply, budget deficit and inflation. As a prerequisite of the VECM model, stationarity properties of the sampled variables in the study are checked to avoid inconsistent and unreliable results. The VECM is applicable when the series are integrated of order one [I (1)] (or at the first difference), and there exists a long run relationship (cointegration) among the variables. In establishing the stationarity properties of the series, the parametric Augmented Dickey-Fuller (ADF) test (Dickey & Fuller, 1979, 1981) and the non-parametric Phillips and Perron (1988) are employed. In these tests, the null hypothesis of unit root (nonstationarity) is tested against the alternative hypothesis of stationarity (no unit root). The rejection (non-rejection) of the null hypothesis implies that the series are stationary (nonstationary) within the sampled period. After establishing the stationarity properties of the series, the cointegration test of the variables is tested using the Johansen cointegration test introduced by Johansen and Juselius (1990), which provides two test statistics-the trace and maximum eigenvalue test statistics—to decide whether there exists a long run relationship among the variables. The null hypothesis of no cointegration is rejected when either the trace or maximum eigenvalues statistic exceeds the 5% significance level. In this study, the optimal lag selection is chosen using the Schwartz Bayesian Criterion (SBC). According to Pesaran and Pesaran (2010), SBC provides more parsimonious specification of the model and more suitable for relatively smaller sample.

After confirmation of a valid long run relationship among the variables, the study first estimates the direction of causality among the variables using the pair-wise Granger causality test. The vector error correction model (VECM), which can produce the long run relationship among the variables, is then estimated after the causality test for a valid confirmation of the relationship as well as the exact impact of the explanatory variables on the dependent variable. Equations (15) and (16) are the general and individual variable specification of the vector error correction model, respectively.

$$\Delta X_t = \prod X_{t-1} + \Gamma_1 \Delta X_{t-1} + \dots + \Gamma_{\rho-1} \Delta X_{t-1+1} + u_t$$
(15)

where  $X_t$  is an m x 1 vector of first difference variables (inflation, money supply, budget deficit, exchange rate, real income and nominal interest rate),  $\Gamma$ 's (1, 2, ...,  $\rho$ -1) and  $\prod$  denote the short-run and long-run parameters of the respective variables.  $u_t$  is the error term such that  $u_t \sim N$  (0,  $\sigma^2$ ). The notation,  $\prod = \alpha \beta'$ ,  $\alpha$  and  $\beta$  are 6 x r matrices that denote the short-run to long-run adjustments coefficients and the cointegration vectors among the variables.

The general form can be expressed in individual variables form as follows:

$$\begin{split} \Delta \ln P_{t} &= \alpha_{0} + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln P_{t-i} + \sum_{i=1}^{k-1} \phi_{i} \Delta \ln BD_{t-i} + \sum_{i=1}^{k-1} \tau_{i} \Delta \ln M_{t-i}^{s} + \sum_{i=1}^{k-1} \omega_{i} \Delta \ln ER_{t-i} + \\ \sum_{i=1}^{k-1} \theta_{i} \Delta \ln Y_{t-i} + \sum_{i=1}^{k-1} \delta_{i} \Delta \ln I_{t-i} + \gamma_{1} ECT_{t-1} + \varepsilon_{1t} \\ \Delta \ln BD_{t} &= \alpha_{0} + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln P_{t-i} + \sum_{i=1}^{k-1} \phi_{i} \Delta \ln BD_{t-i} + \sum_{i=1}^{k-1} \tau_{i} \Delta \ln M_{t-i}^{s} + \sum_{i=1}^{k-1} \omega_{i} \Delta \ln ER_{t-i} + \\ \sum_{i=1}^{k-1} \theta_{i} \Delta \ln Y_{t-i} + \sum_{i=1}^{k-1} \delta_{i} \Delta \ln I_{t-i} + \gamma_{2} ECT_{t-1} + \varepsilon_{2t} \\ \Delta \ln M_{t}^{s} &= + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln P_{t-i} + \sum_{i=1}^{k-1} \phi_{i} \Delta \ln BD_{t-i} + \sum_{i=1}^{k-1} \tau_{i} \Delta \ln M_{t-i}^{s} + \sum_{i=1}^{k-1} \omega_{i} \Delta \ln ER_{t-i} + \\ \sum_{i=1}^{k-1} \theta_{i} \Delta n Y_{t-i} + \sum_{i=1}^{k-1} \delta_{i} \Delta \ln I_{t-i} + \gamma_{3} ECT_{t-1} + \varepsilon_{3t} \\ \Delta \ln ER_{t} &= \alpha_{0} + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln P_{t-i} + \sum_{i=1}^{k-1} \phi_{i} \Delta \ln BD_{t-i} + \sum_{i=1}^{k-1} \tau_{i} \Delta \ln M_{t-i}^{s} + \sum_{i=1}^{k-1} \omega_{i} \Delta \ln ER_{t-i} + \\ \sum_{i=1}^{k-1} \theta_{i} \Delta n Y_{t-i} + \sum_{i=1}^{k-1} \delta_{i} \Delta \ln I_{t-i} + \gamma_{4} ECT_{t-1} + \varepsilon_{4t} \\ \Delta \ln P_{t} &= \alpha_{0} + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln I_{t-i} + \gamma_{5} ECT_{t-1} + \varepsilon_{5t} \\ \Delta \ln I_{t} &= \alpha_{0} + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln I_{t-i} + \gamma_{5} ECT_{t-1} + \varepsilon_{5t} \\ \Delta \ln I_{t} &= \alpha_{0} + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln I_{t-i} + \gamma_{5} ECT_{t-1} + \varepsilon_{5t} \\ \Delta \ln I_{t} &= \alpha_{0} + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln P_{t-i} + \sum_{i=1}^{k-1} \phi_{i} \Delta \ln BD_{t-i} + \sum_{i=1}^{k-1} \tau_{i} \Delta \ln M_{t-i}^{s} + \sum_{i=1}^{k-1} \omega_{i} \Delta \ln ER_{t-i} + \\ \sum_{i=1}^{k-1} \theta_{i} \Delta \ln Y_{t-i} + \sum_{i=1}^{k-1} \delta_{i} \Delta \ln I_{t-i} + \gamma_{5} ECT_{t-1} + \varepsilon_{5t} \\ \Delta \ln I_{t} &= \alpha_{0} + \sum_{i=1}^{k-1} \beta_{i} \Delta \ln P_{t-i} + \sum_{i=1}^{k-1} \phi_{i} \Delta \ln BD_{t-i} + \sum_{i=1}^{k-1} \tau_{i} \Delta \ln M_{t-i}^{s} + \sum_{i=1}^{k-1} \omega_{i} \Delta \ln ER_{t-i} + \\ \sum_{i=1}^{k-1} \theta_{i} \Delta \ln Y_{t-i} + \sum_{i=1}^{k-1} \delta_{i} \Delta \ln I_{t-i} + \gamma_{6} ECT_{t-1} + \varepsilon_{6t} \\ \end{array}$$

where all the variables are as explained earlier,  $\Delta$  represents first difference operator, ECT is the error correction term and  $\gamma$ 's (1, 2, 3, ..., 6) are the short-run coefficients of the error correction term, which lie within 0 and 1 and must be negative and significant. The  $\beta$ ,  $\phi$ ,  $\tau$ ,  $\omega$ ,  $\theta$  and  $\delta$  represent the coefficients of the respective variables.

Furthermore, as indicated earlier in the introduction section regarding the major contribution of the paper, the weak exogeneity test within the VECM framework is employed to assess whether money supply and budget deficit variables used in the study are exogenous to ascertain which of the theories discussed earlier (especially the classical, the monetarist and FTPL theories) hold in the Ghanaian context. The non-rejection of the null hypothesis of the weak exogeneity test implies that the tested variable satisfies weak exogeneity, and therefore the variable is said to be exogenous; otherwise, the variable is endogenous. Also, the study performs series of diagnostic tests to confirm that the estimations are free from any econometric problem. To this end, the normality and autocorrelation issues are checked using the Jarque-Bera and the Lagrange Multiplier test, respectively, while the White heteroscedasticity test is employed to check the issue of heteroscedasticity. In these tests, the null hypothesis of absence of these problems is tested against the alternative hypothesis of the presence of these problems. The non-rejection (rejection) of the null hypothesis indicates the absence (presence) of these problems in the study.

#### 3.3. Data and variable description

The study employs quarterly time series data spanning 1999Q1 to 2019Q4. The sources of data for the study are Bank of Ghana Research Department Database, International Financial Statistics and World Bank's World Development Indicators. Specifically, data on money supply (M1 and M2) and budget deficit is obtained from Bank of Ghana Research Department Database, whereas those on exchange rate and inflation are obtained from International Financial Statistics of the International Monetary Fund (IMF; 2019). Real income and interest rate data is sourced from World Bank's World Development Indicators (World Bank, 2019). Table 1 shows a brief description of the variables used in the study.

Table 1. Description of variables					
Variable	Measurement/proxy	Notation	A priori expectation		
Inflation	Consumer price index	INF	-		
Money supply	Narrow (M1) and broad money (M2)	M1 and M2	Positive		
Budget deficit	Total tax revenue minus total expenditure and net lending	BD	Positive		
Exchange rate	Real effective exchange rate	ER	Positive/Negative		
Real income	Real gross domestic product	Y	Negative		
Interest rate	Deposit interest rate	INT	Positive		

Note: Except for exchange rate and income, all other variables are in nominal terms.

#### 4. Results and discussion

This section discusses the empirical results of the study. The section starts with the summary of descriptive statistics of variables, trends of inflation, budget deficit and money supply (the key variables of the study), and this is followed by the analysis of the stationarity properties of the series. Afterwards, the cointegration test, pair-wise Granger causality test, normalized long run results, weak exogeneity test, impulse response functions and the diagnostic tests are discussed accordingly.

#### 4.1. Descriptive and trend analysis

The summary of the descriptive statistics in terms of mean, standard deviation, skewness, kurtosis, Jarque-Bera and linear correlation are reported in Tables 2 and 3, respectively.

From Table 2, it is observed that money supply (M2), budget deficit (BD) and inflation (INF) have mean (standard deviation) values of 8.90 (1.62), 7.99 (1.59) and 2.63 (0.40) respectively. The maximum (minimum) values for money supply (M2), budget deficit (BD) and inflation (INF) are 11.05 (5.77), 10.18 (5.09) and 3.71 (1.58), respectively. In all, it is observed that the sample variables do not deviate much from their respective mean values as indicated by the standard deviation values. Furthermore, the values for the Skewness, Kurtosis and the Jarque-Bera show that the data is normally distributed. Turning to the linear correlation results (Table 3), it is

Table 2. Summary of descriptive statistics							
	LNINF	LNM1	LNM2	LNBD	LNER	LNY	LNINT
Mean	2.6327	8.2077	8.6997	7.9984	4.5063	-0.9367	2.6164
Median	2.5856	8.2116	8.7549	8.1624	4.5444	-0.8833	2.5777
Maximum	3.7138	10.5544	11.052	10.1798	5.0411	0.5435	3.4859
Minimum	1.7821	5.2954	5.7735	5.0927	4.0564	-2.3758	2.1424
Std. Dev.	0.3961	1.5940	1.6204	1.5855	0.1957	0.7348	0.3428
Skewness	0.6130	-0.1958	-0.2028	-0.2199	0.1406	-0.0252	0.9691
Kurtosis	3.3775	1.8438	1.8197	1.6795	3.2958	2.3931	3.4414
Jarque- Bera	5.7593	5.2153	5.4511	6.7803	0.5831	1.2979	13.831
Probability	0.0562	0.0737	0.0655	0.0537	0.7471	0.5226	0.0610

Note: INF, (M1 and M2), BD, ER, Y, INT denote inflation, money supply, budget deficit, exchange rate, real income, and interest rate respectively.

Table 3. Linear correlation results							
	LNINF	LNM1	LNM2	LNBD	LNER	LNY	LNINT
LNINF	1.0000						
LNM1	-0.3980 (0.0002)	1.0000					
LNM2	-0.3960 (0.0002)	0.9994 (0.0000)	1.0000				
LNBD	-0.3366 (0.0017)	0.9826 (0.0000)	0.9824 (0.0000)	1.0000			
LNER	0.0081 (0.9419)	-0.7428 (0.0000)	-0.7420 (0.0000)	-0.7151 (0.0000)	1.0000		
LNY	-0.8824 (0.0000)	0.3701 (0.0005)	0.3702 (0.0005)	0.3297 (0.0022)	0.0769 (0.4866)	1.0000	
LNINT	0.6178 (0.0000)	-0.3808 (0.0004)	-0.3795 (0.0004)	-0.3678 (0.0006)	-0.0890 (0.4206)	0.6260 (0.0000)	1.0000

Note: In parenthesis is the probability values; INF, (M1 and M2), BD, ER, Y, INT denote inflation, money supply, budget deficit, exchange rate, real income, and interest rate respectively.

Figure 1. Trends of money supply (M1 and M2), budget deficit (BD) and inflation (INF). Source: Authors' construction



Table 4. Unit root test						
Variables	ADF Test		P-P Te		Test	
	I(0)	I(1)		I(0)	I(1)	
lnINF	-3.494**	-5.256***		-2.548	-5.856***	
lnM1	-1.730	-9.780***		-2.602	-15.609***	
lnM2	-2.244	-9.078***		-3.479**	-15.305***	
lnBD	-1.418	-4.660***		-1.472	-5.764***	
lnER	-2.915**	-5.215***		-2.498	-6.720***	
lnY	-2.741	-4.079***		-2.155	-5.311***	
lnINT	-2.317	-3.511**		-2.031	-4.575***	

Note: \*\*\* and \*\* denote significance level at 1 and 5 percent respectively; INF, (M1 and M2), BD, ER, Y, INT denote inflation, money supply, budget deficit, exchange rate, real income, and interest rate, respectively,

At most 5

At most 4 At most 5 3.8415

17.1477

3.8415

0.1462

Trace statistics							
Hypothesized No. of CE(s)	Eigenvalue	Trace statistics	Critical value (0.05)	Prob.**			
None*	0.5175	151.4516	107.3466	0.0000			
At most 1*	0.4354	91.6876	79.3415	0.0044			
At most 2	0.2038	44.8156	55.2458	0.2964			
At most 3	0.1633	26.1307	35.0109	0.3202			
At most 4	0.1084	11.5153	18.3977	0.3464			
At most 5	0.0254	2.1114	3.8415	0.1462			
Maximum eigenvalue	2						
Hypothesized No. of CE(s)	Eigenvalue	Max-Eigen statistic	Critical value (0.05)	Prob.**			
None*	0.5175	59.7641	43.4198	0.0004			
At most 1*	0.4354	46.8719	37.1636	0.0029			
At most 2	0.2038	18.6849	30.8151	0.6562			
At most 3	0.1633	14.6154	24.2520	0.5318			
At most 4	0.1084	9.4040	17.1477	0.4545			

## Table 54 Johansen cointegration test results (Model 1)

Note: \* denotes rejection of the null hypothesis (no cointegration) at 5% significance level and \*\* represents MacKinnon-Haug-Michelis (1999) p-values. Trace and max-eigenvalue tests indicate 2 cointegrating eqn(s) at the 0.05 level.

2.1114

#### Table 5B. Johansen cointegration test results (Model 2) Trace statistics

0.1073

0.0042

0.0254

Hypothesized No. of CE(s)	Eigenvalue	Trace statistics	Critical value (0.05)	Prob.**
None*	0.5182	139.3887	107.3466	0.0001
At most 1*	0.3724	79.5024	79.3415	0.0486
At most 2	0.2232	41.3018	55.2458	0.4542
At most 3	0.1249	20.5920	35.0109	0.6686
At most 4	0.1073	9.6501	18.3977	0.5146
At most 5	0.0042	0.3471	3.8415	0.5557
Maximum Eigenvalue	9			
Hypothesized No. of CE(s)	Eigenvalue	Max-Eigen statistic	Critical value (0.05)	Prob.**
None*	0.5182	59.8864	43.4198	0.0004
At most 1*	0.3724	38.2006	37.1636	0.0379
At most 2	0.2232	20.7098	30.8151	0.4944
At most 3	0.1249	10.9420	24.2520	0.8455

0.3471 Note: \* denotes rejection of the null hypothesis (no cointegration) at 5% significance level and \*\* represents MacKinnon-Haug-Michelis (1999) p-values. Trace and max-eigenvalue tests indicate 2 cointegrating eqn(s) at the 0.05 level.

9.3029

observed that money supply (M1 and M2) and budget deficit (BD) have a weak significant negative correlation with inflation. However, budget deficit (BD) has a strong significant positive correlation with money supply (M1 and M2).

0.4646

0.5557

The trends of money supply (M1 and M2), budget deficit (BD) and inflation (INF) over the study period (1999Q1 to 2019Q4) are shown in Figure 1.

It is observed that money supply (M1 and M2) has a steady upward trend over the study period. However, budget deficit and inflation have been fluctuating over the years, but the former shows an upward trend. It is further observed that, in periods of election (2000, 2004, 2008, 2012 and 2016), both budget deficit and inflation tend to rise, indicating that budget deficit could explain inflation dynamics in Ghana.

#### 4.2. Unit root test

The results from the ADF and P-P unit root tests are reported in Table 4. From the results, both tests confirm that narrow money (M1), budget deficit, real income and interest rate are all stationary at the first difference [I (1)]. However, exchange rate and inflation are stationary at the first difference [I (1)] in the P-P test, whereas the ADF test confirms that broad money (M2) is stationary at the first difference [I (1)]. Following the confirmation of stationarity properties of the variables, the study proceeds with the Johansen cointegration test.

#### 4.3. Johansen cointegration test

The Johansen cointegration (trace and maximum-eigenvalue statistics) results are reported in Table 5A and Table 5B.

The results from the Johansen cointegration test in both estimations (Models 1 and 2) reveal that there exists a long run relationship among the variables. Specifically, the trace and maximumeigenvalue tests for Models 1 and 2 show that there exists at least two cointegrations among the variables at 5% significance level. Given the valid long run relationship among the variables, the study then continues with the direction of causality analysis using the pair-wise Granger causality test, after which the VECM technique is also employed for the long run dynamics.

#### 4.4. Pair-wise granger causality test

The results from the pair-wise Granger causality among the variables is reported in Table 6. The results show the money supply (M1 and M2), budget deficit and real income Granger-cause inflation; however, the null hypothesis of the opposite is not rejected. The result regarding money supply (M1 and M2) causing inflation supports both the classical theory and monetarist hypothesis, which claim that inflation is always and everywhere a monetary phenomenon. Further, the FTPL, which argues that fiscal (budget) deficit causes inflation, also holds. However, to completely ascertain these claims, these theories propose that money supply and budget deficit should be exogenously determined.

Given that, this proposition is not obtained from the Granger causality analysis, it is therefore important to conduct the weak exogeneity test, especially for the validation of the classical, the monetarist and the FTPL theories that explain the relationship among budget (fiscal) deficit, money supply and inflation. Furthermore, the fact that real income (output) and interest rate Granger-cause inflation also supports the structuralist and Keynesian theories regarding inflation determination. It is also revealed that money supply Granger-causes budget deficit but budget deficit does not Granger-causes money supply. This result is, however, contrary to the hypothesis by Sargent and Wallace (1981) that budget-deficit Granger-causes money supply and afterwards, money supply Granger-causes inflation. The latter is observed in this study, but the former does not hold in this context. About interest rate and inflation, the results reveal a bi-directional causal relationship, which reiterate the significance of interest rate in determining inflation.

#### 4.5. Johansen normalized long run results

The results of the normalized long run estimates are reported in Table 7. Narrow money (M1) and broad money (M2) are estimated separately, and the results are reported under Models 1 and 2,

Table 6. Pair-wise Granger causality test results (F-statistics)						
Pair-variables	Causal direction					
	$\Rightarrow$	<del>(</del>				
M1, INF	3.9786**	0.2784				
M2, INF	3.9889**	0.1751				
BD, INF	3.4879**	0.7085				
ER, INF	2.1702	1.3974				
Y, INF	3.2949**	0.7620				
INT, INF	4.6635**	2.6871*				
BD, M1	1.1996	4.7429**				
BD, M2	1.4560	5.1369**				

Note: \*\* and \* denote rejection of the null hypothesis at 5 and 10% significance level, respectively; INF, (M1 and M2), BD, ER, Y, INT denote inflation, money supply, budget deficit, exchange rate, real income, and interest rate, respectively.

#### Table 7. Johansen normalized long run results Model 1

Variable	lnINF	lnM1	lnBD	lnER	lnY	lnINT
INF Equation	1.0000	0.3948*** (0.1185)	-0.2485*** (0.0979)	0.4875* (0.2813)	0.2217*** (0.0482)	-0.1515* (0.0899)
M1 Equation	2.5329*** (0.3107)	1.0000	-0.6294*** (0.0755)	1.2347** (0.5384)	0.5616*** (0.1551)	-0.3837* (0.2199)
BD Equation	-4.0243*** (0.4965)	-1.5888*** (0.1459)	1.0000	-1.9617* (1.0235)	-0.8922*** (0.2491)	0.6097* (0.3584)
Model 2 (Bench	nmark)					
Variable	lnINF	lnM2	lnBD	lnER	lnY	lnINT
INF Equation	1.0000	0.4119*** (0.1209)	-0.2896*** (0.1063)	0.3357 (0.2674)	0.2203*** (0.0496)	-0.1934** (0.0903)
M2 Equation	2.4279*** (0.3079)	1.0000	-0.7030*** (0.0746)	0.8150 (0.5335)	0.5348*** (0.1536)	-0.4696** (0.2179)
BD Equation	-3.4533*** (0.4382)	-1.4224*** (0.1208)	1.0000	-1.1592 (0.8668)	-0.7607*** (0.2210)	0.6680** (0.3144)
Adjustment co	efficients (γ's)					
		Model 1			Model 2	
γ <sub>1</sub>		-0.5104*** (0.0728)			-0.5143*** (0.0726)	
γ <sub>2</sub>		-0.3282*** (0.1091)			-0.3634*** (0.1090)	
$\gamma_3$		-0.3440*** (0.1197)			-0.3456*** (0.1203)	

Note: \*\*\*, \*\* and \* represent 1, 5 and 10% significance levels, respectively. Model 1 is where the estimation is done using narrow money (M1) and Model 2 is when the estimation is done using broad money (M2). Values in the parentheses are the standard errors; INF, (M1 and M2), BD, ER, Y, INT denote inflation, money supply, budget deficit, exchange rate, real income, and interest rate, respectively

respectively. It must be emphasized that in interpreting VECM results, the signs of the coefficients are interpreted in the reverse form—a positive (negative) coefficient means a negative (positive) relationship. The study also uses Model 2 as the benchmark model for the interpretation of the elasticities (coefficient), because, in measuring money supply, M2 captures M1.

Regarding the estimations based on the normalization of inflation and money supply (M1 and M2), the results reveal a significant negative relationship between inflation and money supply (M1 and M2) in both inflation and money supply (M1 and M2) equations. The coefficients (Model 2) indicate that all other things been equal, a 1% increase (decrease) in broad money supply decreases (increases) inflation by approximately 0.41% at 1% significance level, whereas the coefficient of inflation in the money supply equation (Model 2) suggests that a 1% rise (fall) in inflation causes broad money (M2) supply to decrease (increase) by about 2.43% at 1% significance level. The results further show that the negative effect of inflation on money supply is relatively higher than that of money supply on inflation. The negative impact of money supply (M2) on inflation implies that inflation in the context of Ghana is not caused by money supply but could be attributed to other factors (as argued by the Keynesian and Structuralist theories) in the economy as revealed by the study. Thus, this result does not support the classical and monetarist theories that money supply is the cause of inflation. This result contradicts findings of studies on Ghana by Bawumia and Abradu-Otoo (2003), Adu and Marbuah (2011), Adom et al. (2015), Adjei (2018), and Boamah (2019); however, it is consistent (in terms of the negative relationship but not significance) with the study by Kovanen (2011).

The results also indicate that budget deficit (inflation) is revealed to exert a significant positive effect on inflation (budget deficit) in both inflation and budget deficit equations in Models 1 and 2 when inflation and budget deficit are normalized. The coefficient of budget deficit from Model 2 shows that all other factors held constant, a 1% increase (decrease) in budget deficit induces a rise (fall) in inflation by about 0.29% at 1% significance level. Also, a 1% rise (fall) in inflation causes a rise (reduction) in budget deficit by about 3.45% at 1% level of significance.

These results seem realistic, as budget deficit financed by printing of money or borrowing (through issuing of government securities) causes disequilibrium in the money market, which in turn raises the prices of non-tradable goods, and hence a rise in general price level of goods and services (inflation). On the other hand, inflation in an economy implies higher government expenditure because rising prices of goods and services cause the cost of government projects to rise and, therefore, exceeding its revenue, which results in budget deficit. The significant positive impact of the budget deficit validates the FTPL. Our result regarding the effect of budget deficit on inflation contradicts the insignificant findings by Adom et al. (2015) and Adu and Marbuah (2011). However, this finding is consistent with studies on countries in other parts of the world (Anantha Ramu, 2014; Lin & Chu, 2013; Ndanshau, 2012; Nguyen, 2015). The significant positive impact of budget deficit on inflation implies that budget deficit is very crucial in determining inflation in the Ghanaian economy, and hence, policymakers should not overlook its potency.

Furthermore, normalizing money supply (M1 and M2) and budget deficit in both models, the results reveal that money supply (M1 and M2) exerts a significant positive impact on budget deficit, and budget deficit also has a significant positive effect on money supply (see money supply and budget deficit equations) in both Models 1 and 2. From Model 2, the coefficient of budget deficit indicates that money supply will increase (decrease) by about 0.70% if budget deficit increases (decreases) by 1%, whereas the coefficient of money supply shows that a 1% increase (decrease) in money supply increases (reduces) budget deficit by about 1.42%, and these are significant at 1% level. The positive impact of budget deficit on money supply also seems factual, as budget deficit financed by printing of money (seigniorage) or borrowing increases money supply in an economy. Exchange rate is found to exert a negative impact on inflation and money supply, but the effect on budget deficit is positive in both models. However, the impacts in Model 2 are insignificant. The coefficients from Model 1 reveal that a rise in exchange rate (measured by real effective exchange rate) by 1% induces inflation to fall by about 0.49% at 1% significance level. This result is plausible in the sense that when there is a rise in real effective exchange rate, domestic goods become more expensive and less competitive relative to their trading partners. As a result, exports are expected to fall, while imports are expected to increase, all other things being equal. The fall in exports implies that there will be more goods in the domestic market (higher supply), which is likely to

Table 8. Weak exogeneity test results					
	lnM1	lnM2	lnBD		
Model 1	1.1769 (0.2780)		0.0284 (0.8662)		
Model 2		0.5361 (0.4640)	0.1183 (0.7309)		

Note: Probability values are in the parentheses; (M1 and M2) and BD denote money supply and budget deficit, respectively.

exceed the quantity demanded by consumers, and hence a fall in the general price level. This outcome is consistent with the results obtained by Adu and Marbuah (2011) and Anantha Ramu (2014).

Regarding real income (output), the results from both models show that its effect on inflation and money supply (M1 and M2) is significantly negative. However, the effect on budget deficit is positive in both models. This means that expansion in output growth in an economy is considered as a potential way of reducing inflation. The coefficient in Model 2 indicates that a 1% rise (fall) in real income lowers (increases) inflation by about 0.22%, ceteris paribus. Studies by Bawumia and Abradu-Otoo (2003), Adu and Marbuah (2011), Adjei (2018), and Boamah (2019) on Ghana and Anantha Ramu (2014) on India have also reported a significant negative relationship between real income and inflation. The results further show that interest rate exerts a significant positive effect on inflation and money supply, but the effect is negative for budget deficit in both models. This significant positive outcome of the interest rate on inflation is consistent with previous studies on Ghana (see: Adom et al., 2015; Adu & Marbuah, 2011; Boamah, 2019).

From Table 7, the adjustment coefficients ( $\gamma$ 's) confirm a valid cointegration among the variables. Specifically, disequilibrium in inflation in the short run will be restored to its long run equilibrium at a speed of approximately 51%. In addition, the significant negative coefficients indicate the stability of the models.

#### 4.6. The weak exogeneity test

The results from the weak exogeneity test (specifically for money supply and budget deficit) are reported in Table 8.

It is observed that both money supply and budget (fiscal) deficit satisfy the weak exogeneity test. This is because the study fails to reject the null hypothesis of weak exogeneity (exogenous) since the probability values of both money supply (M1 and M2) and budget deficit (BD) are all greater than the 5% significance level. This implies that money supply and budget deficit are exogenous or are exogenously determined statistically.

Figure 2a. Response of inflation to money supply and budget deficit shocks. Source: Authors' construction



Response to Cholesky One S.D. Innovations Response to Cholesky One S.D. Innovations

Figure 2b. Response of money supply to inflation and budget deficit shocks. Source: Authors' construction





#### Response to Cholesky One S.D. Innovations



Figure 2c. Response of budget deficit to money supply and inflation shocks. Source: Authors' construction

#### Response to Cholesky One S.D. Innovations



#### Response to Cholesky One S.D. Innovations



Table 9. Diagnostic test results of the VECM model						
	Model 1	Model 2				
Diagnostic test	Test statistic	Test statistic				
Serial correlation LM test	28.3992 (0.8127)	36.4435 (0.4480)				
White heteroscedasticity	13.4608 (0.4906)	13.2099 (0.5101)				
Normality	5.2153 (0.0737)	5.4511 (0.0655)				

Note: Probability values are in the parentheses.

#### 4.7. Analysis of the impulse response function

To further examine the dynamics of money supply, budget deficit and inflation, we go a step further to analyze how inflation responds to budget deficit and money supply shocks. We also analyze how money supply (budget deficit) responds to budget deficit and inflation (money supply [M2] and inflation) shocks. The impulse response functions are presented in Figures 2a-2c.

Starting with Figure 2a (response of inflation to money supply [M2] and budget deficit shocks), it is observed that inflation responds negatively to money supply (M2) shocks over the period (see left panel). Specifically, the response declines with marginal fluctuations from period 1 to period 6 and remains stable from periods 6 to 10 with a marginal increase in period 9. The implication is that shocks to money supply reduces inflation. Again, the response of inflation to budget deficit shocks (see right panel) indicates that inflation responds more positively to budget deficit shocks (especially from period 4 to period 10). However, the response is negative from the 1<sup>st</sup> period to 3<sup>rd</sup> period. Therefore, the implication is that shocks to the budget deficit increase inflation, given the more positive response to budget-deficit shocks.

Regarding the response of money supply (M2) to inflation and budget deficit shocks (Figure 2b), the results show that money supply (M2) responds negatively to inflation over the period (see left

panel). Precisely, the response declines from period 1 to period 5 and then with a little rise from period 6 to period 8. It remains relatively stable for the rest of the periods. This implies that a shock to inflation does not increase money supply (M2) in the Ghanaian economy. In addition, it is indicated that money supply (M2) responds positively to budget deficit shocks (from period 1 to period 10) (see the right panel). Given the positive response, a shock to the budget deficit has a higher likelihood of increasing the money supply (M2) in the Ghanaian economy.

Furthermore, Figure 2c indicates that the response of the budget deficit to the shocks of the money supply (M2) is negative and declines from period 1 to period 3 and remains relatively stable throughout the remaining periods (see left panel). The negative response implies that a shock to money supply (M2) in the economy reduces budget deficit. On the other hand, the response of budget deficit to inflation shocks is positive over the period. The response from period 1 to period 10 is stable with a marginal rise in periods 3 and 5. The positive response suggests that any shock to inflation will increase budget deficit in Ghana.

#### 4.8. Diagnostic test

In ensuring the outcome from the estimation is robust, reliable and good for effective policy purposes, various diagnostic tests (normality, serial correlation, and heteroscedasticity) are performed, and the results are reported in Table 9. It is observed that both models are free from the aforementioned econometric problems. This is because the probability values of the serial correlation, heteroscedasticity and normality tests exceed 5% significance level, implying non-rejection of the null hypothesis of serial correlation, heteroscedasticity and normality tests.

#### 5. Concluding remarks and policy implications

Using quarterly data over the period 1999Q1 to 2019Q4, this paper has explored the relationship (dynamics) between budget (fiscal) deficit, money supply and inflation in Ghana. Also, using the weak exogeneity test, the study tests which of the theories (especially the classical, the monetarist and FTPL theories) explain the relationship between inflation, money supply and budget deficit in the Ghanaian context. The pair-wise Granger-causality test as well as the vector error correction model (VECM) are also employed for the analysis. The results from the pair-wise Granger causality test show that there is a unidirectional causal relationship that runs from money supply, budget deficit, and real income to inflation. There is also a unidirectional causal relationship that moves from money supply to budget deficit. The results further indicate that there is a bi-directional causal relationship between interest rate and inflation. Regarding the VECM analysis, the benchmark results (Model 2) reveal that money supply has significant negative relationship with inflation, whereas budget deficit is found to exert a significant positive effect on inflation. It is further revealed that money supply and inflation have significant positive relationship with budget deficit. The budget deficit also exerts a significant positive effect on the money supply. The effect of exchange rate on inflation is also negative albeit insignificant. Furthermore, the effects of real income and interest rate on inflation, money supply and budget deficit are revealed to be significant. The results from the impulse response function also indicate that inflation responds more positively to budget deficit. However, it (inflation) tends to respond negatively to money supply (M2) shocks. Also, budget deficit responds positively (negatively) to inflation (money supply [M2]) shocks. Further, money supply responds more positively (negatively) to budget deficit (inflation) shocks. Based on the VECM, the weak exogeneity test and impulse response function, it is concluded that the fiscal theory of the price level (FTPL) holds in the Ghanaian context as it is able to explain the nexus between money supply, budget deficit and inflation. The classical and monetarist theories, however, do not hold in the Ghanaian context.

The findings of the study have some policy implications for government of Ghana and other African countries that share similar characteristics with Ghana. Based on the positive effect of the budget deficit on inflation, the study suggests that a reduction in government spending is one sure way to curb inflation, as this minimizes government expenditure. The study therefore recommends reduction in government spending. Specifically, this can be achieved by reducing spending on unproductive sectors of the economy, reducing the size of the government, and putting stringent measures in place to control the

bureaucratic nature of government officials. Further, there should also be conscientious effort by the government to ensure spending within the allocated budget. This will ensure that expenditure of the government is within its budget to avoid or reduce budget deficit. Another implication of the findings is that, in curbing inflation, policymakers should pay more attention to budget deficit rather than money supply. In all, ensuring reduction in government expenditure, government size and avoiding budget deficit is likely to result in stable and favorable inflation in the Ghanaian economy, which is also likely to enhance economic growth and development.

However, there is a caveat in generalizing the findings of the present study to other African countries as the data used relate to Ghana, and this becomes a limitation of the study. Given this, future studies should consider using a panel of African countries to improve the generalization of the results and validate the present findings.

#### Author details

Emmanuel Duodu<sup>1,2</sup> E-mail: emmanuelduodu9@gmail.com ORCID ID: http://orcid.org/0000-0003-4988-2043 Samuel Tawiah Baidoo<sup>1,3</sup> E-mail: samueltawiahbaidoo@yahoo.com ORCID ID: http://orcid.org/0000-0001-5941-6308 Hadrat Yusif<sup>1</sup> ORCID ID: http://orcid.org/0000-0002-6316-4016 Prince Boakye Frimpong<sup>1</sup> E-mail: pbfrimpong@gmail.com

- ORCID ID: http://orcid.org/0000-0002-2829-9257 <sup>1</sup> Department of Economics, Kwame Nkrumah University
- of Science and Technology, Kumasi Ghana. <sup>2</sup> Department of Fundamentals of Economic Analysis,
- University of Alicante, Alicante Spain. <sup>3</sup> Department of Accounting and Finance, School of Business, Christian Service University College, Kumasi Ghana

#### correction

This article has been republished with minor changes. These changes do not impact the academic content of the article.

#### **Disclosure statement**

No potential conflict of interest was reported by the author(s).

#### Funding

The authors received no direct funding for this research.

#### **Citation information**

Cite this article as: Money supply, budget deficit and inflation dynamics in Ghana: An empirical investigation, Emmanuel Duodu, Samuel Tawiah Baidoo, Hadrat Yusif & Prince Boakye Frimpong, *Cogent Business & Management* (2022), 9: 2043810.

#### References

- Adjei, S. K. (2018). Inflation determinants Milton Friedman's theory and the evidence from Ghana, 1965-2012 (Using ARDL Framework). International Journal of Applied Economics, Finance and Accounting, 3(1), 21–36. https://doi.org/10.33094/8.2017.2018.31.21.36
- Adom, P. K., Agradi, M. P., & Quaidoo, C. (2018). The transition probabilities for inflation episodes in Ghana. International Journal of Emerging Markets, 13 (6), 2028–2046. https://doi.org/10.1108/IJoEM-08-2017-0313
- Adom, P. K., Zumah, F., Mubarik, A. W., Ntodi, A. B., & Darko, C. N. (2015). Analysing inflation dynamics in Ghana. African Development Review, 27(1), 1–13. https://doi.org/10.1111/1467-8268.12118
- Adu, G., & Marbuah, G. (2011). Determinants of inflation in Ghana: An empirical investigation. South African

Journal of Economics, 79(3), 251–269. https://doi.org/ 10.1111/j.1813-6982.2011.01273.x

- Ahiabor, G. (2013). The effects of monetary policy on inflation in Ghana. Developing Country Studies, 3(12), 82–89.
- Amoah, L., & Aziakpono, M. J. (2018). Exchange rate pass-through to consumer prices in Ghana: Is there asymmetry? International Journal of Emerging Markets, 13(1), 162–184. https://doi.org/10.1108/ IJoEM-07-2016-0179
- Anantha Ramu, M. R. (2014). Relationship between fiscal deficit and inflation in India: A long-term empirical analysis. *The Indian Economic Journal*, 62(3), 1099–1120. https://doi.org/10.1177/ 0019466220140303
- Arndt, H. W. (1985). The origins of structuralism. World Development, 13(2), 151–159. https://doi.org/10. 1016/0305-750X(85)90001-4
- Bawumia, M., & Abradu-Otoo, P. (2003). Monetary growth, exchange rates and inflation in Ghana: An error correction analysis. Bank of Ghana (Working Paper). Bank of Ghana, Accra.
- Boamah, M. I. (2019). Inflation Dynamics in a Small Developing Economy: An Empirical Analysis for Ghana. The Journal of Developing Areas, 53(3), 229–237. https://doi.org/10.1353/jda.2019.0049
- Chenery, H. B. (1975). The structuralist approach to development policy. *The American Economic Review*, 65(2), 310–316 Accessed 13 January 2021 https:// www.jstor.org/stable/1818870.
- Chiaraah, A., & Nkegbe, P. K. (2014). GDP growth, money growth, exchange rate and inflation in Ghana. Journal of Contemporary Issues in Business Research, 3(2), 75–87 Accessed 13 January 2021 http://cite seerx.ist.psu.edu/viewdoc/download?doi=10.1.1.680. 6400&rep=rep1&type=pdf.
- Dickey, D. A., & Fuller, W. A. (1979). Distribution of the estimators for autoregressive time series with a unit root. Journal of the American Statistical Association, 74(366a), 427–431 https://doi.org/10.1080/ 01621459.1979.10482531.
- Dickey, D. A., & Fuller, W. A. (1981). Likelihood ratio statistics for autoregressive time series with a unit root. *Econometrica: Journal of the Econometric Society*, 49 (4), 1057–1072. https://doi.org/10.2307/1912517
- Friedman, M. (1963). Inflation: Causes and consequences. Asia Publishing House.
- Frisch, H. (1989) Inflation Theories). Elif Printing Limited, Istanbul, Turkey: .
- Ghartey, E. E. (2001). Macroeconomic instability and inflationary financing in Ghana. *Economic Modelling*, 18(3), 415–433. https://doi.org/10.1016/S0264-9993(00)00047-X
- Granger, C. W. (1969). Investigating causal relations by econometric models and cross-spectral methods.

Econometrica: Journal of the Econometric Society, 37 (3), 424–438. https://doi.org/10.2307/1912791 Gyebi, F., & Boafo, G. K. (2013). Macroeconomic determi-

nants of inflation in Ghana from 1990-2009. International Journal of Business and Social Research (IJBSR), 3(6), 81–93 Accessed 13 January 2021 https://citeseerx.ist.psu.edu/viewdoc/download?doi= 10.1.1.683.3210&rep=rep1&type=pdf.

International Monetary Fund (IMF). (2019). International Financial Statistics (IFS).

- Johansen, S. (1988). Statistical analysis of cointegration vectors. Journal of Economic Dynamics & Control, 12 (2), 231–254. https://doi.org/10.1016/0165-1889(88) 90041-3
- Johansen, S., & Juselius, K. (1990). Maximum likelihood estimation and inference on cointegration-with applications to the demand for money. Oxford Bulletin of Economics and Statistics, 52(2), 169–210. https://doi.org/10.1111/j.1468-0084.1990. mp52002003.x

Kotwal, O. P. (1987). Theories of Inflation: A Critical Survey. The McGraw-Hill Publishing Companies.

- Kovanen, A. (2011). IMF Working Papers (London, United Kingdom: Palgrave Macmillan), 1–23.
- Laryea, S. A., & Sumaila, U. R. (2001). Determinants of inflation in Tanzania. Chr. Michelsen Institute.
- Lin, H. Y., & Chu, H. P. (2013). Are fiscal deficits inflationary? Journal of International Money and Finance, 32(1), 214–233. https://doi.org/10.1016/j. jimonfin.2012.04.006
- Lozano-Espitia, L. I., & Lozano-Espitia, I. (2008). Budget deficit, money growth and inflation: Evidence from the Colombian case. Borradores de Economía, Central Bank of Colombia, Colombia, No. 537 1-25 Accessed 13 January 2021 https://repositorio.banrep. gov.co/bitstream/handle/20.500.12134/5554/ BORRADOR%20537.pdf?sequence=1.
- Nasir, M. A., Balsalobre-Lorente, D., & Huynh, T. L. D. (2020a). Anchoring inflation expectations in the face of oil shocks & in the proximity of ZLB: A tale of two targeters. *Energy Economics*, 86, 104662. https://doi.org/10.1016/j.eneco.2020. 104662
- Nasir, M. A., Huynh, T. L. D., & Vo, X. V. (2020b). Exchange rate pass-through & management of inflation expectations in a small open inflation targeting economy. International Review of Economics & Finance, 69, 178–188. https://doi.org/10.1016/j.iref. 2020.04.010
- Ndanshau, M. O. (2012). Budget deficits, money supply and inflation in Tanzania: A multivariate granger causality test, 1967–2010. University of Dar Es Salaam Working Paper, 04(12), 1–36 doi:http://dx.doi. org/10.2139/ssrn.2142328).
- Nguyen, B. (2015). Effects of fiscal deficit and money M2 supply on inflation: Evidence from selected

economies of Asia. Journal of Economics, Finance and Administrative Science, 20(1), 49–53. https://doi.org/ 10.1016/j.jefas.2015.01.002

- Noyola, V. J. (1956). El desarollo economico y la inflacion en Mexico y otros paises latino-americanos. *Investagacion Economica*, 16(4), 603–649.
- Olivera, J. H. (1964). On Structural Inflation and Latin-American'Structuralism'. Oxford Economic Papers, 16 (3), 321–332. https://doi.org/10.1093/oxfordjournals. oep.a040958
- Pesaran, B., & Pesaran, M. H. (2010). Time series econometrics using Microfit 5.0: A user's manual. Oxford University Press, Inc. https://pdfs.semanticscholar. org/8773/45b2c11ce28d55a1a15d6de3078fbfa75f1f. pdf
- Phillips, P. C., & Perron, P. (1988). Testing for a unit root in time series regression. *Biometrika*, 75(2), 335–346. https://doi.org/10.1093/biomet/75.2.335
- Sargent, T. J., & Wallace, N. (1981). Some unpleasant monetarist arithmetic. Federal Reserve Bank of Minneapolis Quarterly Review, 5(3), 1–17 https:// d1wqtxts1xzle7.cloudfront.net/35204863/QR531with-cover-page-v2.pdf?Expires= 1646000868&Signature=SIShuw5JuJ-u~Uefo4gXP~

SauhYjLL4gUquOtBAN2KKvtItsVGBG8sB0GemK52R8D yWnT7szbelU47hn~

0PCvxGRrGJPC9bQMH8CeOu0miknk-FnnmuIkROT8DtgRms8qihd06WmBxpGWEsiZwZj-Hbxt5Y266f1eKe7m3bW9C6Leu6bIhm~

CoTHUxIm2cYulO8k~9YWDMVSsa7Q9vJehlSiDS-2DHTwz-

4BYkeIM5ICIP8t9pQq7hi6RjPg5qcAFnca2xVMdzfp W1~ee8ecX9sqLtUfg-WLq4Nw~iNoMt9iW1ccccz/d1E64XDapXppWpgPApcz1bcEPgl

iNoMt9iW1scgcz4dIkS4XDnnYppWpaRAeszIhcERqllIs VYQ\_&Key-Pair-Id=APKAJLOHF5GGSLRBV4ZA.

- Sowa, N. K. (1994). Fiscal deficits, output growth and inflation targets in Ghana. World Development, 22(8), 1105–1117. https://doi.org/10.1016/0305-750X(94) 90079-5
- Sunkel, O. (1958). La inflación chilena: Un enfoque heterodoxo. El Trimestre Económico, 25(4), 570–599 https://www.jstor.org/stable/pdf/20855451.pdf?casa\_ token=UDthiri0JTUAAAAA:

WVfGtFfaBrN0rEyJPJ3xYyCbE0zdQlXmiDS\_ QxS3iQ4cboeMlHUvfgmMot-XpWvYSncJqXK4vsH\_ FXxBNREbU3DsM76Pdpi2bF2okbYS4Nqfjklxtxbi.

- Ubide, M. A. J. (1997). Determinants of inflation in Mozambique. Working Paper, No. 97-145. International Monetary Fund.
- Woodford, M. (1995). Price-level determinacy without control of a monetary aggregate. In *Carnegie-Rochester conference series on public policy*, Vol.43, pp.1–46, North Holland.
- World Bank. (2019). World Banks' World Development Indicators.



#### © 2022 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

You are free to:

Share — copy and redistribute the material in any medium or format. Adapt — remix, transform, and build upon the material for any purpose, even commercially. The licensor cannot revoke these freedoms as long as you follow the license terms. Under the following terms: Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

## *Cogent Business & Management* (ISSN: 2331-1975) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- Discounts and waivers for authors in developing regions

Submit your manuscript to a Cogent OA journal at www.CogentOA.com